Author’s response to reviews

Title: Access to healthcare and alternative health-seeking strategies among undocumented migrants in Denmark

Authors:

Dan Biswas (dabi@sund.ku.dk)
Maria Kristiansen (makk@sund.ku.dk)
Allan Krasnik (alk@sund.ku.dk)
Marie Norredam (mano@sund.ku.dk)

Version: 8 Date: 6 July 2011

Author’s response to reviews: see over
Authors’ response to reviews

Regarding the manuscript: 1620093397504496 - Access to healthcare and alternative health-seeking strategies among undocumented migrants in Denmark

Authors:

Dan Biswas: dabi@sund.ku.dk
Maria Kristiansen: makk@sund.ku.dk
Allan Krasnik: alk@sund.ku.dk
Marie Norredam: mano@sund.ku.dk

Version: 6
1) Reference 6 has been included in the main manuscript and all references are now in numerical order. Moreover, the references formerly cited in figure 1 are now cited within the manuscript.

2) As discussed with Mr Philip Dooner from the The BMC Series Editorial Production Team, figures are to be included within the manuscript and not as appendix.

Yours sincerely
Dan Biswas